Author's response to reviews

Title: Do the mutations of C1GALT1C1 gene play important roles in the genetic susceptibility to Chinese IgA nephropathy?

Authors:

Gui-Sen Li (guisenli@163.com)
Guang-Jun Nie (nienie8888@yahoo.com.cn)
Hong Zhang (hongzh@bjmu.edu.cn)
Ji-Cheng Lv (chenglv@263.net)
Yan Shen (sheny@chgb.org.cn)
Hai-Yan Wang (why@bjmu.edu.cn)

Version: 6 Date: 15 September 2009

Author's response to reviews:

Dear editors,

Our manuscripts titled "Do the mutations of C1GALT1C1 gene play important roles in the genetic susceptibility to Chinese IgA nephropathy?" (MS: 4437889032345027) was checked carefully by us and uploaded again. These changes were all small typographical changes which were marked in red words. We hoped it would be suitable for publication in your journal.

Thank you for attentions.

Best regards.

Guisen Li